Author’s response to reviews

Title: Overcoming Bcr-Abl T315I mutation by combination of GNF-2 and ATP competitors in an Abl-independent mechanism.

Authors:

Mamduh Khateb (smmdoh@gmail.com)
Nili Ruimi (nilir@migal.org.il)
Hazem Khamisie (khazem25@yahoo.com)
Yousef Najajre (ynajajreh@pharm.alquds.edu)
Afsar Mian (mian@med.uni-frankfurt.de)
Anna Metodieva (metodiev@med.uni-frankfurt.de)
Martin Ruthardt (ruthardt@med.uni-frankfurt.de)
Jamal Mahajna (jamalm@migal.org.il)

Version: 10 Date: 22 October 2012

Author’s response to reviews: see over
Dear Editor,

October 22, 2012

List of Changes:

1. A serious of experiments were carried to assess the ability of GNF-2 and AKIs to cooperate in affecting proliferation and clonigenicity of the parental Ba/F3 cells. Data are summarized in the supplemental materials (Figure S1-S3).
2. In Page 8, a sentence describing the cooperation of GNF-2 and AKIs in controlling proliferation of the parental Ba/F3 was added. (in Red)
3. In page 9, a sentence describing the cooperation of GNF-2 and AKIs in controlling clonigenicity of the parental Ba/F3 was added. (in Red)
4. In page 15, a sentence summarizing the lack of cooperation between GNF-2 and AKIs in controlling proliferation and clonigenicity of the parental Ba/F3 was added. (in Red)
5. Details of volume of pages were added to Ref #14 (Page 20).

Best wishes,
Jamal Mahajna, Phd.